Dear Editors,

Enclosed you find the revised manuscript entitled “Hepatic tuberculosis presenting with extremely high serum ferritin masquerading as Adult onset Still’s disease - A Case report” is herewith submitted to you for presentation in “Journal of Medical Case Reports”. All comments and recommendations of the editorial office were followed in the current version of the manuscript.

The Associate Editor of JMCR handling our submission has passed on the following comment:

1. We request that your manuscript be checked over by a native English speaker and the language revised.

- mistakes and language errors of the manuscript was corrected by a native
english speaker. All the changes made were highlighted while revising the manuscript to make it easier for the Editors to go for a prompt decision on our manuscript.

2. Please remove the figure containing the image of the patients face.
   - Removed

3. Please remove the sentence; "unexpected or unusual presentations of a disease ," as this not needed for publication.
   - Removed

4. Please also use the term 'male' rather than 'man' or 'businessman'.
   - Amended

5. In the Abstract; "We report a case of hepatic TB, presented with clinical criteria of AOSD and extreme hyperferritinemia, which paused a diagnostic confusion." ... should be ...
   "We report a case of hepatic TB, presented with clinical criteria of AOSD and extreme hyperferritinemia, which caused a diagnostic confusion."
   - Amended

Regards,

E.M. Manoj